Author’s response to reviews

Title: The presence of a booster phenomenon among contacts of active pulmonary tuberculosis cases: a retrospective cohort

Authors:

Cristiane G Salles (Cristiane.SALLES@br.netgrs.com)
Antonio Ruffino-Netto (aruffino@fmrp.usp.br)
Jose R Lapa-e-Silva (jrlapa.ntg@terra.com.br)
Afranio L Kritski (kritski@hucff.ufri.br)
Michelle Cailleaux-Cesar (mi_cailleaux@yahoo.com.br)
Fernanda C Queiroz-Mello (fcqmello@hucff.ufri.br)
Marcus B Conde (marcusconde@hucff.ufri.br)

Version: 5 Date: 15 March 2007

Author’s response to reviews:

Thank you for the message. Please, find the revised version uploaded.